Author’s response to reviews

Title: Disentangling the Genetics of Sarcopenia: prioritization of NUDT3 and KLF5 as genes for lean mass & HLA-DQB1-AS1 for hand grip strength with the associated enhancing SNPs & a Scoring system

Authors:
Abhishek Singh (abhishek.narain@iitdalumni.com)

Version: 4 Date: 11 Feb 2020

Author’s response to reviews:

Dear Reviewers

The response points are stated in order as it was asked:

1. Re-formatted the abstract to include Methods section as well in addition to background, results and conclusions. Added 'Keywords' section.

2. Made changes in Ethics approval and consent to participate section - 'Not Applicable'

3. Tables were used to prepare the Figures for the Venn Diagrams. They have been appropriately stated in the main body in the results section.

4. Removed previous attachments.